Author's response to reviews

Title: Explanations of socioeconomic differences in changes in physical function in older adults: results from the Longitudinal Aging Study Amsterdam

Authors:

Annemarie Koster (a.koster@zw.unimaas.nl)
Hans Bosma (hans.bosma@zw.unimaas.nl)
Marjolein I. Broese van Groenou (mi.broese@fsw.vu.nl)
Gertrudis I.J.M. Kempen (g.kempen@zw.unimaas.nl)
Brenda W.J.H. Penninx (Brendap@ggzba.nl)
Jacques TH.M. van Eijk (j.vaneijk@zw.unimaas.nl)
Dorly J.H. Deeg (djh.deeg@vumc.nl)

Version: 6 Date: 5 October 2006

Dear Editor,

My manuscript is currently in the proofreading stage. I carefully checked my manuscript for any small typographical changes I would like to make. There were a few minor changes I made on page 8.

line 4: replaced 'is' by 'was'
line 6: replaced 'includes' by 'included'
line 7: removed the first 'the'
line 8: replaced 'develops' by 'developed' and 'includes' by 'included'

Kind regards,
Annemarie Koster